Author's response to reviews

Title: Artemisia annua L.: Sesquiterpene lactone fraction antinociceptive activity

Authors:

Fabricio de Faveri Favero (fabricioffavero@gmail.com)
Rogério Grando (roggrando2@yahoo.com.br)
Fabiana R Nonato (frnonato@uol.com.br)
Giovanna B Longato (giovannabl@yahoo.com.br)
Ilza MO Sousa (ilza@cpqba.unicamp.br)
Núbia CA Queiroz (nubia.cassia@hotmail.com)
João E Carvalho (carvalho_je@yahoo.com.br)
Humberto M Spindola (hmspindola@bol.com.br)
Mary A Foglio (foglioma@gmail.com)

Version: 2  Date: 24 February 2014

Author's response to reviews:

Dear Sir,

Enclosed is the manuscript with title “Artemisia annua L.: Evidence of Sesquiterpene lactones’ fraction antinociceptive activity “concerning the antinociceptive activity of the enriched sesquiterpene lactone fraction to be submitted as research article. The author(s) declare that they have no competing interests.

The authors’ contributions were:

FFF: made substantial contributions to conception and acquisition of data, analysis and interpretation of data; was involved in drafting the manuscript or revising it critically for important intellectual content

RG: made substantial contributions to acquisition of data, analysis and interpretation of data

FRN; made substantial contributions to acquisition of data, analysis and interpretation of data

GBL; made substantial contributions to acquisition of data, analysis and interpretation of data

IMOS; made substantial contributions to acquisition of data, analysis and interpretation of data

NCAQ: made substantial contributions to acquisition of data, analysis and interpretation of data

JEC: Gave final approval of the version to be published

HMS: made substantial contributions to conception and acquisition of data, analysis and interpretation of data; was involved in drafting the manuscript or revising it critically for important intellectual content.
MAF: made substantial contributions to conception and acquisition of data, analysis and interpretation of data; was involved in drafting the manuscript or revising it critically for important intellectual content. Gave final approval of the version to be published; and agrees to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Hoping the data presented herein meet the high standards of Plos One.

Best regards,

Mary Ann Foglio